Author’s response to reviews

Title: Pulmonary embolism causing atrial fibrillation with slow ventricular response: a case report

Authors:

Mohammad Abdulsalam (msalam87@hotmail.com)

Mohammed Elganainy (mohammednabil_eg@hotmail.com)

Ahmad Abdulsalam (dr.ahmad.j.abdulsalam@gmail.com)

Version: 2 Date: 12 Feb 2019

Author's response to reviews:

Dear Editor,

We thank you for your reply. Regarding the changes that have been suggested, we present our reply:

Reviewer #1 Reply:

1- The word sino-atrial node was replaced by atrio-ventricular node.

2- The time between onset of AF with slow ventricular response and ventricular implantation was added to the manuscript.

3- We agree regarding with the reviewer regarding a more watchful approach, however the patient was hemodynamically unstable.

4- More details were added to the follow up data.
Reviewer #2 Reply:

1- The word Doppler was edited.
2- The basal diameter was edited.
3- A short comment was added in the discussion describing McConnell’s sign and a reference of Lopez-Candales was added.

We hope that it will be accepted for publication.

Thank you for your diligence.

Kind Regards,

Ahmad J Abdulsalam